Author’s response to reviews

Title: Protection against ultraviolet radiation by commercial summer clothing: need for standardised testing and labelling

Authors:

   Thilo Gambichler (t.gambichler@derma.de)
   Sebastian Rotterdam (s.rotterdam@derma.de)
   Peter Altmeyer (p.altmeyer@derma.de)
   Klaus Hoffmann (k.hoffmann@derma.de)

Version: 2 Date: 19 Oct 2001

In short, all recommendations respective corrections of the referees have been included in the paper.